Author’s response to reviews

Title: Does housework keep you healthy? The contribution of domestic physical activity to meeting current recommendations.

Authors:

Marie H Murphy (mh.murphy@ulster.ac.uk)
Paul Donnelly (pauldonnelly@sportni.net)
Gavin Breslin (g.breslin1@ulster.ac.uk)
Simon Shibli (S.Shibli@shu.ac.uk)
Alan M Nevill (a.m.nevill@wlv.ac.uk)

Version: 4 Date: 15 January 2013

Author’s response to reviews:

An ethics statement has been added to the methods.
A conflict of interest has been added to the end of the paper
A statement of author contributions has been added to the end of the paper